Fri Oct 2 16:33:28 GMVIr 1998
Year: 98 Doy: 275
Qobserver: yasukawa

Fri Oct 2 16:34:21 GVI 1998 CH P Startup--Initializing new tape
WEATHER COMMENT: Fri Cct 2 16:34:38 GMI 1998
Cool, clear, light south w nd.

**CH P PROBLEM *: Fri Oct 2 16:34:52 GMI 1998

CH P filter temp OK this norning at startup, 34.488.

Fri Oct 2 16:42:41 GV 1998 CH P CH P Start Patro
**CH P PROBLEM *: Fri COct 2 16:42:56 GMTI 1998

Once sun warmed filter, tenmp went into red , presently 34.691
Fri Oct 2 16:43:53 GV 1998 MKI | Start Patro

**CH P PROBLEM *: Fri COct 2 16:56:52 GMI 1998

Filter tenp. now 35. 607

Fri Ot 2 17:00:49 GMI 1998 CHP Bi as

Fri Gt 2 17:01:44 GMI 1998 CH P End Bi as

Fri Ot 2 17:01:57 GMI 1998 CH P Wat er

Fri Cct 2 17:02:42 GMI 1998 Pl CS Fl at

Fri Ot 2 17:02:40 GMI 1998 CHP End Water
Fri Ot 2 17:04:09 GMI 1998 Pl CS End Fl at

**PSPT PROBLEMF*: Fri Cct 2 17:18:27 GMr 1998

PSPT started up OK, brushed front surface of entrance filter.
Active mrror stable. Jeff Kuhn is supposed to arrive shortly
to investigate active mirror problem

MKI Il COWMMENT: Fri Oct 2 17:20:07 GVII 1998

CORRECT started up inadvertently. | rnust have brushed the nouse
while cursor was sitting on button at sone point.

**CH P PROBLEMF™*: Fri Cct 2 17:22:33 GMIr 1998

tenp 37.103

Fri Oct 2 17:24:59 GJMI 1998 MKI | Start Patro

MKI Il COWMMENT: Fri Oct 2 17:25:44 GV 1998

CORRECT was done a bit early but we were supposed to do one this nonth.
It is a good day, weather-wise, to do it.

**CH P PROBLEMF™*: Fri Cct 2 17:32:09 GVr 1998

tenp 37.598

**CH P PROBLEM*: Fri Oct 2 17:46:21 GVIT 1998
38. 255

Fri Cct 2 18:02:48 GMI 1998 Pl CS Fl at

Fri Cct 2 18:02:49 GMI 1998 CH P Bi as

Fri Cct 2 18:03:46 GMI 1998 CH P End Bi as

**PSPT PROBLEMF*: Fri COct 2 18:03:59 GMI 1998
1800 UT wake-up OK, active mirror stable.

ri Oct 2 18:03:59 GV 1998 CH P Wat er
**CH P PROBLEM*: Fri COct 2 18:04:29 GMI 1998
38.979



Fri Cct 2 18:04:54 GMI 1998 CH P End Wat er
**PSPT PROBLEMF*: Fri Oct 2 18:20:55 GVII 1998
1800 UT run went OK, active mrror was stable.
**CH P PROBLEM*: Fri Oct 2 18:21:44 GVII 1998

39. 594

**CH P PROBLEM*: Fri COct 2 18:35:33 GMI' 1998
40. 063

Fri Oct 2 19:02:46 GV 1998 PI CS Fl at

Fri Oct 2 19:02:56 GMI 1998 CH P Bi as

Fri Cct 2 19:04:01 GVr 1998 CH P End Bi as
Fri Cct 2 19:04:01 GVIr 1998 CH P End Bi as
Fri Oct 2 19:04:15 GV 1998 CH P Wat er

**PSPT PROBLEM*: Fri COct 2 19:04:44 GVI 1998

1900 UT Run is oscillating. Jeff Kuhn arrived a few m nutes before
and is now investigating the problem

**CH P PROBLEM*: Fri Oct 2 19:05:43 GVI 1998

40. 907

COMMENT: Fri Oct 2 19:11:03 GMI 1998

St oppi ng to reconfigure donme shutter.

Fri Oct 2 19:11:37 GV 1998 PI CS End Patr ol

Fri Oct 2 19:11:31 GMIr 1998 CH P CH P End Patr ol
Fri Cct 2 19:17:02 GMVIr 1998 MKI T | Start Patrol
Fri Cct 2 19:19:30 GVIr 1998 Pl CS Start Patrol

Fri Oct 2 19:19:30 GMIr 1998 Pl CS Start Patrol

COWMENT: Fri Qct 2 19:27:41 GVI 1998

Net work i s down.

**CH P PROBLEMF*: Fri Cct 2 19:28:16 GMIr 1998
41. 414

WEATHER COWMMENT: Fri Cct 2 19:28:45 GMI 1998
Sky is very bright near sun, noisy.

COWMMENT: Fri Qct 2 19:50:45 GVI 1998

Net back up.

Fri Oct 2 20:01:56 GMI 1998 CH P Gin

Fri Oct 2 20:02:42 GMI 1998 Pl CS Fl at
CH P COWENT: Fri Oct 2 20:03:07 GVI 1998

42. 056

Fri Cct 2 20:04:08 GMI 1998 Pl CS End FIl at
Fri OQct 2 20:06:37 GMI 1998 CH P End Gain
Fri Oct 2 20:06:51 GMI 1998 CH P Bi as

Fri Oct 2 20:07:44 GMI 1998 CHP End Bi as
Fri Ot 2 20:07:54 GMI 1998 CHP Wat er

Fri Oct 2 20:08:33 GMI 1998 CHP End Water
**CH P PROBLEM*: Fri Oct 2 20:15:48 GVII 1998
42. 260

setpoint remains at .642V

sensor voltage is slowy oscillating between .699V-. 733V
**PSPT PROBLEMF*: Fri Cct 2 20:17:12 GMr 1998

Jeff suspected sonething flaky on the Mrror Drive board,



board we first suspected. There may be a bad connection

somewhere as we can get the mirror servo neters to oscillate by
jogging the cardd while it is out on extenders. Jeff is resoldering
all of the connections on the board.

**CH P PROBLEMF*: Fri Cct 2 20:27:41 GMr 1998

42.419 and still rising.

setpoint at .642V

sensor oscillating between .701V and . 747V

Turned tenp. controller off, getting concerned that the tenp

is getting too high.

**CH P PROBLEMF*: Fri Cct 2 20:32:58 GMIr 1998

St oppi ng CHI P.

Fri Oct 2 20:35:19 GVr 1998 CH P CH P End Patro

**CH P PROBLEM™*: Fri Cct 2 20:57:34 GV 1998

temp coning down, 37.999 and dropping slowy. Controller off.

Fri Oct 2 21:02:40 GVr 1998 Pl CS Fl at

Fri Oct 2 21:04:10 GMVr 1998 Pl CS End Fl at

**CH P PROBLEMF*: Fri Cct 2 21:05:32 GMr 1998

Turned control |l er back on.

37.504 when | did that, Tenmp is still dropping, now at 37104.
**CH P PROBLEM*: Fri COct 2 21:14:53 GMI 1998

Tenmp is going back up. Turning controller back off to protect filter.
WEATHER COMMVENT: Fri Cct 2 21:42:23 GMr 1998

I n stratocunul us cl ouds.

**PSPT PROBLEMF*: Fri Cct 2 21:44:15 GV 1998

Jeff and | agree that the mirror drive board is flaky. Resoldering
did not fix problem | think there is a hairline flaw in the power
or ground circuit as this is the only circuit commn to both

X and Y drive circuits. Tenperature and vibration cause the

drive circuits to peg between limts. The reason | got off that track
the last tine when | was cooling the board with the fan was the air
pressure was causing the board to waver to and fro in the breeze.
Thi s mechani cal vibration was al so causing the board to get fl aky.
Ther mal expansi on probably caused fl akyness when the board was
securely nounted in the HOT chassis. W currently have the board
out on extenders and tied down with a wire. PSPT is running OK
Jeff Kuhn has left satisfied that we resol ved the probl em

Haosheng will send out a spare mrror drive board to replace the

fl aky one.

Fri Oct 2 21:52:03 GMVI 1998 CH P endi ng tape

PSPT COMMENT: Fri Cct 2 21:53:42 GV 1998

Optics around active mrror and beamsplitter dusted off.

Fri Oct 2 22:02:45 GMVr 1998 Pl CS Fl at

Fri Oct 2 22:04:13 GV 1998 PI CS End FI at

COMMENT: Fri Oct 2 22:07:05 GMI 1998

Tapes:
PICS: P01310



CH P: C00707
LOAL:  LO0545

Fri Oct 2 22:08:34 GMIr 1998
Fri Cct 2 22:14:28 GMVI 1998

MKI T

16_43. rawrk3
16_47. rawrk3
16_49. rawmrk3
16 _52. rawnk3
16_55. rawnk3
16_58. rawmrk3
17_01. rawrk3
17_04. r awmrk3
17_07. rawrk3
17 _10. rawnk3
17 _13. rawnk3
17_16. rawrk3
17_25. rawrk3
17_28. rawrk3
17_31. rawrk3
17 33. rawnk3
17 _36. rawnk3
17_39. rawrk3
17_42. rawrk3
17_45. r awmrk3
17_48. r awmrk3
17 51. rawnk3

17_54.
17_57.
18_03.
18_09.
18_15.
18_18.
18_21.
18 24.
18 27.
18_30.
18_33.
18_36.
18_39.
18~ 42.
18_45.
18_47.
18_50.
18_53.
18_56.
18_509.
19_02.
19_05.

rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3

PI CS

19_08.
19 17.
19_20.
19 22.
19_25.
19_28.
19_31.
19_34.
19_37.
19_40.
19_43.
19_46.
19_49.
19_52.
19_55.
19_58.
20_00.
20_03.
20_06.
20_09.
20 12.
20_15.

End Patr ol
r awnk3 20_18.
r awmrk3 20_21.
r awmrk3 20_24.
rawmrk 3 20 27.
rawmrk 3 20_30.
rawnk3 20_33.
r awnk3 20_36.
r awmrk3 20_38.
r awmrk3 20_41.
rawmrk 3 20 _44.
rawmrk 3 20 _47.
rawnk3 20_50.
r awnk3 20_53.
r awmrk3 20_56.
r awmrk3 20_59.
rawmrk 3 21 02.
rawmrk 3 21 05.
r awnk3 21 08.
r awnk3 21 11.
r awmrk3 21_13.
r awmrk3 21_16
rawmrk 3 21 19

rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3

21 22.rawrk3
21_25. rawmrk3
21_28. rawrk3
21 31.rawnk3
21 34.rawk3
21 37.rawrk3
21 40.rawrk3
21_43. rawrk3
21_46. r amrk3
21 49. rawk3
21 51. rawk3
21 54.rawrk3
21 57.rawrk3
22_00. r awmrk3
22_03. rawrk3
c18 _00.rawrk3
c18_06.rawrk3
cl1l8 12.rawnk3



